Author’s response to reviews

Title: Characteristics and resource needs in patients with vestibular symptoms: a comparison of patients with symptoms of unknown versus determined origin

Authors:

Martin Müller (martin.mueller2@insel.ch)
Martina Goeldlin (martinabeatrice.goeldlin@insel.ch)
Janika Gaschen (janika.gaschen@bluewin.ch)
Thomas C. Sauter (Thomas.Sauter@insel.ch)
Stephanie Stock (stephanie.stock@uk-koeln.de)
Franca Wagner (franca.wagner@insel.ch)
Aristomenis K. Exadaktylos (aristomenis.exadaktylos@insel.ch)
Urs Fischer (urs.fischer@insel.ch)
Roger Kalla (roger.kalla@insel.ch)
Georgios Mantokoudis (georgios.mantokoudis@insel.ch)

Version: 2 Date: 09 Aug 2020

Author’s response to reviews:

Dear editor

Thank you for contacting us regarding the order of authors in our manuscript. We had to change the order of authors in the submission form due to a bug in the submission system (which we have reported to the editorial office and was obviously fixed in the mean time). However, the order of authors within the submitted manuscript file has never and will not change.

We however had to change the corresponding author in the manuscript for consistency, as I (Martina Goeldlin, the submitting author) was not able to submit the manuscript (for technical reasons again) unless being corresponding author. All authors, including the originally designed corresponding author, Dr. Martin Müller, have agreed to that change.

Please find the signed form attached.
May we kindly ask you to change the name "Martin Mueller" to "Martin Müller" as indicated in the form, as well?

At this stage, we have omitted all track change files as required.

Kind regards
Martina Goeldlin